Author's response to reviews

Title: Portal Vein Thrombosis; Risk factors, Clinical presentation and Treatment

Authors:

Kirstine K Sogaard (kks@dce.au.dk)
Lone B Astrup (loast@as.aaa.dk)
Hendrik Vilstrup (hvils@as.aaa.dk)
Henning Gronbaek (henning.gronbaek@dadlnet.dk)

Version: 5 Date: 1 August 2007

Author's response to reviews:

Dear Dr Annabel Philips

I have one supplemental comment and correction to the coverletter and manuscript sent on 31 July 2007.

In the methods section in the sentence regarding statistical analysis:
"Significance was calculated with Pearson's Chisquared test."

In this sentence it has been added that Students t-test was used, and therefore in the resubmitted ms it is:
"Significance was calculated with Student's t-test and Chi squared test."